Author's response to reviews

Title: Studying feasibility and effects of a two-stage nursing staff training in residential geriatric care using a 30 month mixed-methods design [ISRCTN24344776]

Authors:

Elsbeth Betschon (elsbeth.betschon@fhsg.ch)
Michael Brach (michael.brach@uni-muenster.de)
Virpi Hantikainen (virpi.hantikainen@fhsg.ch)

Version: 5 Date: 30 May 2011

Author's response to reviews:

Dear Editor,

there were five small typing errors:

p2 (Abstract; Methods/Design)
four subsequent days and an additional counselling day ("day" instead of "days")
[this was also corrected in the abstract form]

p6 (2nd paragraph)
SOPMAS score signifies a decline ("means" was deleted)

p10 (Intervention and measurement points)
on the staff level and on the resident level ("and" instead of "an")

p13 (Discussion)
on the effects of Kinaesthetics ("of" was inserted)

p14 (Authors' contributions)
authors monogram "VP" changed to "VH"

Kind regards,
Michael Brach